Metformin in non-diabetic hyperglycaemia: the GLINT feasibility RCT

Simon J Griffin,^{1,2}* M Angelyn Bethel,³ Rury R Holman,³ Kamlesh Khunti,⁴ Nicholas Wareham,¹ Gwen Brierley,¹ Melanie Davies,⁴ Andrew Dymond,¹ Rose Eichenberger,¹ Philip Evans,⁵ Alastair Gray,⁶ Colin Greaves,⁵ Kyla Harrington,⁴ Graham Hitman,⁷ Greg Irving,² Sarah Lessels,³ Ann Millward,⁵ John R Petrie,⁸ Martin Rutter,⁹ Mike Sampson,^{10,11} Naveed Sattar⁸ and Stephen Sharp¹

- ¹Medical Research Council Epidemiology Unit, University of Cambridge, Cambridge, UK
- ²Institute of Public Health, University of Cambridge, Cambridge, UK
- ³Diabetes Trials Unit, University of Oxford, Oxford, UK
- ⁴Diabetes Research Centre, University of Leicester, Leicester, UK
- ⁵University of Exeter Medical School, Exeter, UK
- ⁶Health Economics Research Centre, Nuffield Department of Population Health, University of Oxford, Oxford, UK
- ⁷Blizard Institute, Barts and The London School of Medicine and Dentistry, Queen Mary University of London, London, UK
- ⁸Institute of Cardiovascular and Medical Sciences, University of Glasgow, Glasgow, UK
- ⁹Faculty of Medical and Human Sciences, University of Manchester, Manchester, UK ¹⁰Norwich Medical School, University of East Anglia, Norwich, UK
- ¹¹Department of Diabetes, Endocrinology and General Medicine, Norfolk and Norwich University Hospitals NHS Foundation Trust, Norwich, UK

*Corresponding author ProfGP@medschl.cam.ac.uk

Declared competing interests of authors: Kamlesh Khunti received honoraria and research grants from AstraZeneca plc, Novartis Pharmaceuticals UK Ltd, Novo Nordisk Ltd, Sanofi S.A., Eli Lilly and Company, Merck Sharp & Dohme Corp., Janssen Research and Development, Boehringer Ingelheim Ltd and Roche Diagnostics Ltd. He has acted as an advisor to the NHS Health Check programme, National Institute for Health and Care Excellence (NICE) Guidance and the Diabetes Prevention Programme External Reference Group. M Angelyn Bethel reports non-financial support from Merck Serono during the conduct of the study. Outside the submitted work, she received grants from Merck Sharp & Dohme Corp. and AstraZeneca, acted as an advisory board member for AstraZeneca, Boehringer Ingelheim and Novo Nordisk, as a speaker for Sanofi and received non-financial support from Bayer plc. Rury R Holman reports ACE (Acarbose Cardiovascular Evaluation) trial funding and lecture fees from Bayer, trial funding from AstraZeneca and Merck Sharp & Dohme Corp. and lecture fees from Boehringer Ingelheim, outside the submitted work. He acted as an advisory board member for Merck, Sharp & Dohme, Elcelyx Therapeutics, Inc., Novartis Pharmaceuticals UK, Novo Nordisk and Servier Laboratories Ltd. Melanie Davies has acted as a consultant, advisory board member and speaker for Novo Nordisk, Sanofi, Eli Lilly and Company, Merck Sharp & Dohme Corp., Boehringer Ingelheim, AstraZeneca and Janssen Research and Development, as an advisory board member for Servier Laboratories and as a speaker for Mitsubishi Tanabe Pharma Corporation and Takeda Pharmaceuticals International, Inc. She has received grants in support of investigator and investigator-initiated trials from Novo Nordisk, Sanofi, Eli Lilly and Company, Boehringer Ingelheim and Janssen Research and Development. She has acted as an advisor to the NHS Health Check programme and the NICE Guidance and is Director of a National Institute for Health Research Biomedical Centre and a Clinical Trials Unit. John R Petrie reports personal fees from Novo Nordisk, Eli Lilly and Company, Janssen Research and Development and Boehringer Ingelheim, grants, personal fees and non-financial support from Sanofi, non-financial support from Merck KGaA and Itamar Medical Ltd, grants from the Juvenile Diabetes Research Foundation Ltd and grants and personal fees from Quintiles (now IQVIA), outside the submitted work. Martin Rutter reports grants from Novo Nordisk, outside the submitted work. He owns stocks and shares in GlaxoSmithKline plc and acted as an advisory board member for Roche Diagnostics Ltd. Naveed Sattar reports personal fees from Boehringer Ingelheim, Janssen Research and Development, Novo Nordisk and Eli Lilly and Company and grants and personal fees from AstraZeneca, outside the submitted work. Simon J Griffin reports receiving an honorarium and reimbursement of travel expenses from Eli Lilly and Company, associated with membership of an Independent Data Monitoring Committee for a randomised trial of a medication to lower glucose levels, and an honorarium from Janssen Research and Development for speaking at an educational meeting in 2015.

Published April 2018 DOI: 10.3310/hta22180

Plain English summary

The GLINT feasibility RCT

Health Technology Assessment 2018; Vol. 22: No. 18 DOI: 10.3310/hta22180

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Cardiovascular disease (CVD), such as a heart attack or a stroke, is the most common long-term condition in the UK. People with diabetes have an increased CVD risk. This risk is also increased in people with raised blood glucose levels who do not have diabetes, a condition referred to as non-diabetic hyperglycaemia (NDH).

Metformin is a cheap and effective diabetes treatment. In people with diabetes, metformin can reduce CVD risk and it may reduce the risk of cancer. It is unknown if metformin has this effect in people with NDH. We designed a trial of metformin (Glucophage[®] SR, Merck KGaA, Bedfont Cross, Middlesex, UK) compared with placebo in people with NDH to answer this question.

First, we evaluated whether or not such a trial was feasible. We tested if people with NDH were willing to take part. We assessed whether or not it was acceptable to post the study drug to participants' homes. We asked participants and their general practitioners (GPs) to complete health questionnaires to see if this data collection method worked. We also asked participants to provide blood samples.

We invited 5251 people to participate in the trial, recruited from general practices and research databases. Just under 10% of those contacted responded and 49% of those met the inclusion criteria. A total of 249 participants took part: 125 received metformin and 124 received an identical-looking placebo tablet.

Posting the study drug was acceptable; only a few packs needed to be replaced and people understood the instructions. The questionnaire return rates were high: 74% of participants and 79% of GPs returned the end-of-study questionnaire.

Most participants were able to take three tablets per day, but around 30% of participants in both the metformin group and the placebo group had stopped taking the study drug by 6 months. Both groups reported side effects, but more participants in the placebo group ranked these as very bothersome. Blood test results indicated that metformin was safe.

Around 20,000 people are needed for the full trial. This would require the use of large patient databases and changes to the trial design.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.236

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 09/01/48. The contractual start date was in June 2013. The draft report began editorial review in January 2017 and was accepted for publication in November 2017. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2018. This work was produced by Griffin *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Health Technology Assessment Editor-in-Chief

Professor Hywel Williams Director, HTA Programme, UK and Foundation Professor and Co-Director of the Centre of Evidence-Based Dermatology, University of Nottingham, UK

NIHR Journals Library Editor-in-Chief

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the EME Programme, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Director of the NIHR Dissemination Centre, University of Southampton, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Director, Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk